Author's response to reviews

Title: Impact of renal dysfunction on long-term outcomes of elderly patients with acute coronary syndrome: a longitudinal, prospective observational study

Authors:

yuqi liu (ametufo4290@gmail.com)
lei gao (gaolei@163.com)
qiao xue (xueq@126.com)
muyang yan (muyangy@126.com)
yang li (yangl@126.com)
pu chen (Puchen@163.com)
yu wang (wangy@163.com)

Version: 6 Date: 4 December 2013

Author's response to reviews: see over
Dear editors:

Thank you for your letter. And I had revised the article according to reviewer 2’s advice and also added limitation part in the article according to answer the reviewer 1’s question. Finally, it is really a very long time since I first submitted this article to your journal. During this time, I had did revised the article according to two reviewer’s advice. Because of the time requirements of my subject, I hope to get your reply of the final decision about the article before the end 2013. Thank you for your help sincerely.

Best regards

Yuqi Liu